



ร sciendo

Sara Nikolova¹*^(D), Daniela Ristikj-Stomnaroska^{1,2}^(D)

¹General Medicine Study Program, University "Goce Delchev", Faculty of Medical Sciences, Shtip, Republic of North Macedonia; ²Department of Neurology, City General Hospital 8th September, Skopje, Republic of North Macedonia

Abstract

Edited by: Igor Spiroski Citation: Nikolova S, Ristiki-Stommaroska D. Transthyretin-related Familial Amyloid Polyneuropathy: A Case Report. Open Access Maced J Med Sci. 2024 Sep 15; 12(3):424-428. https://doi.org/10.3889/camjms.2024.11956 Keywords: Hereditary ATTR amyloidosis; Heterozygous mutations; Transthyretin gene; Case report *Correspondence: Sara Nikolova, University 'Goce Delchev', Faculty of Medical Sciences, Ship, Republic of North Macedonia. E-mail: saranikolova2001@yahoo.com Received: 20-Jul-2024 Revised: 10-Aug-2024 Accepted: 01-Sep-2024 Copyright: © 2024 Sara Nikolova, Daniela Ristiki-Stomnaroska Funding: This research was financially supported by the Scientific Foundation SPIROSKI, Skopje, Republic of North Macedonia Competing Interests: The authors have declared that no competing Interests. The such caccess article distributed under the terms of the Creative Commons Attribution.

NonCommercial 4.0 International License (CC BY-NC 4.0)

BACKGROUND: Hereditary amyloidosis transthyretin is an autosomal dominant disease caused by heterozygous mutations in the transthyretin gene. The disease is characterized by amyloid deposits in various organs, primarily in the peripheral nerves and the myocardium.

CASE PRESENTATION: A 53-year-old female patient with the onset of symptoms 2 years earlier, presented with fatigue, difficulty walking, progressive muscle weakness, tingling in the hands and feet, blood pressure variations, weight loss, and constipation. There was no positive family history of familial amyloid polyneuropathy (FAP). Electromyography revealed sensorimotor axonal neuropathy; electrophoresis of cerebrospinal fluid was of transudative type, without immune activity in the central nervous system, while echocardiography detects hypertrophic myocardium and interventricular septum.

CONCLUSION: All patients who show symptoms of peripheral neuropathy with predominant signs of autonomic nervous system damage and hypertrophic cardiomyopathy should be referred for genetic testing for FAP.

Introduction

Familial amyloid polyneuropathies (FAPs) are a group of life-threatening multisystem disorders transmitted as an autosomal dominant trait [1]. Hereditary transthyretin familial amyloid polyneuropathy (TTR-FAP) is a rare, but severe form of hereditary peripheral neuropathy in adults. It is the most disabling hereditary neuropathy affecting sensory, motor, and autonomic nerves, and is irreversible and fatal within 7–12 years of onset in the absence of therapy [2].

TTR is a secreted polypeptide chain consisting of 127 amino acid residues with an approximate mass of 14 kDa and a prominent β -sheet secondary structure. Mutations in the TTR gene, located on the 18th chromosome, can be associated with hereditary forms of amyloidosis transthyretin (ATTR) that are transmitted as an autosomal-dominant trait [3].

The amyloid accumulation in the peripheral nerves causes distal symmetrical polyneuropathy with neuropathic pain and sensory loss, autonomic dysfunction (e.g., fainting, gastrointestinal symptoms), and motor impairment [4]. As of today, reports of about 100 different points of single or double mutations or of a deletion in the TTR gene have appeared; most of these cases involve small kindreds or no family history [5].

mjms

Case Presentation

We present a 53-year-old female patient with fatigue, difficulty walking, progressive muscle weakness, and tingling in the hands and feet. The first signs appeared 2 years ago, and since then the symptomatology continued to develop gradually. Significant worsening of the symptoms is noted 1 year ago following a serious case of virosis when the tingling feeling in the fingers of both feet and hands gradually spread to the elbows and the middle of both upper legs. The findings are followed with blood pressure variations, constipation, and continuous weight loss (15 kg in 2 years).

The previous medical records reveal a diagnosed hypertrophic cardiomyopathy, two ectopic pregnancies, and splenectomy, and from the family history, we note a father diagnosed with Parkinson's disease and a mother deceased due to colorectal carcinoma. It is important to emphasize that this patient has no family history of FAP.

During the neurological examination, we note a generalized reduced gross motor strength, predominantly of the proximal musculature of the lower limbs, with the presence of hypotrophy of the muscles above and below the knee. The Gowers sign (a sign that presents with any condition that is associated with weakness of the pelvic girdle or proximal muscles of lower extremities [6]) was positive, with present peripheral paresthesia, paraparetic walking, not possible on heel and toes. Furthermore, vegetative symptomatology with gastroparesis, disturbed gastrointestinal tract motility, and blood pressure variations were also present.

The routine biochemical tests (complete blood count, glycemia, degradation products, electrolyte status, enzyme, and lipid status) showed no abnormal findings. The urine was cloudy, with leukocytes present in sediment, negative for nitrites. The thyroid hormones and tumor markers (AFP, CEA, Ca 15-3, Ca 19-9, Ca 125, and HCG) were also within reference ranges. We conducted a protein electrophoresis of cerebrospinal fluid which was of transudative type, without immune activity in the central nervous system.

Electromyography examination shows spontaneous denervation activity of the type of fibrillation potentials in m. genioglossus, m. tibialis anterior dexter, m. quadriceps femoris sinister, m. gastrocnemius sinister, and fasciculations in m. abductor digiti minimi dexter and m. gastrocnemius sinister. The electroneurography examination of the motor nerves registered prolonged latencies with slightly to moderately decreased motor conduction velocities for n. peroneus profundus bilaterally and n. tibialis bill (Table 1). The elecrtoneurography investigation of the sensory nerves registered unmeasurable sensory conduction velocities (SCV) for n. surrealism bilaterally and prolonged latency with reduced SCV for n. medianus sinister (Table 2).

Table 1: Prolonged latencies and decreased MCV for n. peroneus profundus bilaterally and n. tibialis bilaterally

R/L	Nerve	Latency (ms)	MCV
L	n. medianus	4.8	55.0
L	n. ulnaris	2.8	50.0
L	n. peroneus profundus	4.4	32.6
R	n. peroneus profundus	4.0	38.0
R	n. tibialis	5.6	36.0
L	n. tibialis	7.2	35.3
MCV: Motor	conduction velocities		

Particularly important is the results from the conducted imaging examinations. The echocardiographic examination showed concentric hypertrophy of the left ventricle (LV) myocardium with no visible outbursts in myocardial kinetics, good systolic function, and EF%:

Table 2: Prolonged latency with reduced SCV for n. medianus sinister. n. surrealism registered unmeasurable SCV

R/L	Nerve	Latency (ms)	SCV	
L	n. medianus	3.9	41.0	
L	n. ulnaris	2.4	58.3	
R	n. surplus		Unmeasurable	
SCV: Sensory conduction velocities.				

Open Access Maced J Med Sci. 2024 Sep 15; 12(3):424-428.

70%. Interventricular septum significantly hypertrophied, altered with nodular substrates present in the tissue. The SPECT scan of the myocardium with 99mTc-PYP showed intense and diffuse pathological cardiac accumulation of the radiotracer in the LV, a finding indicative of cardiac amyloidosis.

Furthermore, the patient was sent to Macedonian Academy for Sciences and Arts (MANU) for a genetic testing where a pathogenic variant 325G>C (Glu109Gln) was detected, after which the final diagnosis of TTR-related FAP was made.

The patient started therapy with tafamidis 61 mg daily, duloxetin 60 mg once a day every morning, and pregabalin 75 mg daily. Duloxetin and pregabalin were implemented for neuropathic pain relief. The advice of a cardiologist: The patient continued her regular therapy with Aspirin 100 mg, Losartan 50 mg, and Presolol 50 mg daily. A recommendation for physical and rehabilitation treatment was also given.

Discussion

Hereditary amyloidosis TTR-related is a rare disease associated with several point mutations in the TTR gene, determining abnormal aggregations of the TTR protein in different organs [7]. The natural course of TTR-FAP can be classified into three Stages: I (sensory polyneuropathy), II (progressive walking disability), and III (wheelchair-bound or bedridden) [8], with a life expectancy ranging from 7 to 12 years from onset.

As a rare disease, the European prevalence of amyloidosis (including secondary amyloidosis) was estimated at 47/100,000 in 2014 [9]. The most common TTR mutations worldwide are Val30Met, Val122IIe, and Glu89GIn [10]. An endemic cluster of cases has been occurring in Macedonia in recent decades. All patients and carriers are from the region of eastern Macedonia. Dominant cases are Delchevo, Berovo, Vinica, and Strumica. The most common pathogenic mutation in the TTP gene registered is Glu109GIn.

Since it is an autosomal dominant hereditary mutation in the TTR gene, the family history and the screening is extremely important in families with FAP. Genetic testing and counseling for the families are available today for the family members of patients with FAP at the MANU.

TTR-FAP is a highly heterogeneous disease and for its early diagnosis it is important to pay attention to the so-called red flag symptoms: TTR-FAP should be suspected if progressive peripheral sensory-motor neuropathy is observed in combination with one or more of the following: family history of a neuropathy, autonomic dysfunction, cardiac hypertrophy, gastrointestinal problems, inexplicable weight loss,



Figure 1: Diagnostic pathway of ATTR and possible differential diagnoses. Mainly we differentiate ATTR from AL Amyloidosis by analyzing the presence of light chains of immunoglobulins in serum and urine, protein electrophoresis and analyzing urine for proteinuria

carpal tunnel syndrome, renal impairment (albuminuria), or ocular involvement (vitreous opacities) [11].

In a patient with hypertrophic cardiomyopathy and peripheral neuropathy, we can think of several types of amyloidosis: Systemic amyloidosis, ATTR type amyloidosis (which can occur as wild type or hereditary form) and some of the other rare forms including: Apoprotein A1 FAP, Gelsolin FAP, Reactive AA amyloidosis, β 2 microglobulin associated amyloidosis, and senile systemic amyloidosis (Figure 1).

For further differentiation, we approach the analysis of the presence of (1) light chains of immunoglobulins (kappa and lambda) in serum and urine; (2) protein electrophoresis; and (3) urinalysis for proteinuria.

If we get a normal finding, as in our case, then we raise a suspicion of ATTR, which we confirm with 99m Tc-PYR SPECT and a biopsy of the gingiva, gastric, rectal mucosa, fatty tissue or biopsy of the nerve-n surplus with Congo red staining and finally with Genetic testing of the gene for TTR on chromosome 18 for existing mutations, which is performed at MANU.

On the other hand, if we note light chains of immunoglobulins (kappa and lambda) in serum and urine, positive protein electrophoresis, and proteinuria, we continue the investigations in the direction of systemic type amyloidosis with a bone marrow biopsy and consultation with hematology.

The treatment options for TTR-related FAP are limited. Liver transplantation that replaces the variant TTR by the donor wild-type TTR was the first proposed treatment as a procedure that improves survival, especially in patients under 50 years of age with the Val30Met variant [12]. However, liver transplantation rarely improves nerve function, has no effects on manifestations such as eye lesions or cardiac involvement, is limited by the availability of a donor's liver, and requires lifetime immunosuppressant therapy,

which may in turn result in adverse clinical outcomes (e.g. increased risk of infections) [13]. Today, the most promising treatment option for patients is implementing an oral TTR stabilizer (tafamidis) which should be started as soon as the diagnosis is established.

Tafamidis is a first-in-class therapy that slows the progression of TTR amyloidogenesis by stabilizing the mutant TTR tetramer, thereby preventing its dissociation into monomers and amyloidogenic and toxic intermediates [14]. However, even though the drug slows down, it does not prevent the progression of the disease. A 10-year clinical study conducted by Merlini et al. [15] evaluated the effects of tafamidis on mortality in Val30Met and non-Val30Met patients with ATTR-PN. During the 10-year extension study, patients once a day orally administrate 20 mg tafamidis meglumine. Tafamidis has been shown to delay neurological disease progression in ATTR-PN and more recently to reduce the risk of mortality in TTR amyloidosis with cardiomyopathy. Furthermore, the median duration from disease onset to death reported in the literature is ~ 10 years (although this can vary considerably based on genotype and phenotype), whereas in the present study, 75-85% of patients treated with tafamidis are still alive ~ 8-9 years after starting tafamidis treatment, which is estimated as~11-14 years after disease onset [15].

Some of the other promising treatments include Patisiran and Inotersen, RNA inhibitors which prevent the synthesis of TTR and by reducing the concentration of circulating TTR slow down the progression of neuropathy [16]. There are still ongoing clinical trials for implementing human monoclonal antibodies in the treatment of ATTR polyneuropathy. Antibody binding mediates the elimination of ATTR aggregates through phagocytes, which potentially leads to amyloid clearance [17]. Two drugs in this category have reached the clinical development pipeline, both designed to inhibit fibril formation through specifically targeting misfolded TTR, namely PRX004 and NI006 [18].

Conclusion

Identification of ATTR amyloid polyneuropathy can be challenging, especially in non-endemic regions. Patients may present with heterogeneous symptoms often leading to a misdiagnosis of diabetic neuropathy or chronic inflammatory demyelinating polyradiculoneuropathy. Thus, all patients who have symptoms of peripheral neuropathy with predominant signs of autonomic nervous system damage and hypertrophic cardiomyopathy should be referred for genetic testing for FAP. A timely and accurate diagnosis of ATTR amyloidosis allows for early treatment and potentially modifies disease progression in patients.

References

1. Planté-Bordeneuve V, Said G. Familial amyloid polyneuropathy. Lancet Neurol. 2011;10(12):1086-97. https://doi.org/10.1016/ S1474-4422(11)70246-0

PMid:22094129

 Adams D, Beaudonnet G, Adam C, Lacroix C, Théaudin M, Cauquil C, *et al.* Familial amyloid polyneuropathy: When does it stop to be asymptomatic and need a treatment?. Rev Neurol. 2016;172(10):645-52. https://doi.org/10.1016/j. neurol.2016.08.007
PMid:27663057

 Nuvolone M, Obici L, Merlini G. Transthyretin-associated familial amyloid polyneuropathy-current and emerging therapies. US Neurol. 2012;8:24-32. https://doi.org/10.5167/uzh-69213

- Escolano-Lozano F, Barreiros AP, Birklein F, Geber C. Transthyretin familial amyloid polyneuropathy (TTR-FAP): Parameters for early diagnosis. Brain Behav. 2018;8(1):e00889. https://doi.org/10.1002/brb3.889
 PMid:29568686
- Ando Y, Nakamura M, Araki S. Transthyretin-related familial amyloidotic polyneuropathy. Arch Neurol. 2005;62(7):1057-62. https://doi.org/10.1001/archneur.62.7.1057
 PMid:16009758
- Shrestha S, Munakomi S. Gower sign. In: StatPearls. Treasure Island, FL: StatPearls Publishing; 2024. Available from: https:// www.ncbi.nlm.nih.gov/books/NBK540973 [Last accessed on 2023 Feb 13].
- Dugo K, Bruno F, Sturiale V, Brancato D, Saccone S, Federico C. Hereditary transthyretin-related amyloidosis: Genetic heterogeneity and early personalized gene therapy. Biomedicines. 2022;10(10):2394. https://doi.org/10.3390/ biomedicines10102394
 PMid:36289657
- Takei YI, Ikeda SI, Ikegami T, Hashikura Y, Miyagawa SI, Ando Y. Ten years of experience with liver transplantation for familial amyloid polyneuropathy in Japan: Outcomes of living donor liver transplantations. Intern Med. 2005;44(11):1151-6. https://doi.org/10.2169/internalmedicine.44.1151
 PMid:16357452
- Parman Y, Adams D, Obici L, Galán L, Guergueltcheva V, Suhr OB, et al. Sixty years of transthyretin familial amyloid polyneuropathy (TTR-FAP) in Europe: Where are we now? A European network approach to defining the epidemiology and management patterns for TTR-FAP. Curr Opin Neurol. 2016;29(Suppl 1):S3-13. https://doi.org/10.1097/ WCO.00000000000288

PMid:26734951

- Obi CA, Mostertz WC, Griffin JM, Judge DP. ATTR epidemiology, genetics, and prognostic factors. Methodist Debakey Cardiovasc J. 2022;18(2):17-26. https://doi.org/10.14797/mdcvj.1066
 PMid:35414855
- Conceição I, González-Duarte A, Obici L, Schmidt HH, Simoneau D, Ong ML, *et al.* "Red-flag" symptom clusters in transthyretin familial amyloid polyneuropathy. J Peripher Nerv Syst. 2016;21(1):5-9. https://doi.org/10.1111/jns.12153 PMid:26663427
- Planté-Bordeneuve V, Gorram F, Salhi H, Nordine T, Ayache SS, Le Corvoisier P, *et al.* Long-term treatment of transthyretin familial amyloid polyneuropathy with tafamidis: A clinical and neurophysiological study. J Neurol. 2017;264:268-76. https:// doi./org/10.1007/s00415-016-8337-3 PMid:27878441
- 13. Scott LJ. Tafamidis: A review of its use in familial amyloid

polyneuropathy. Drugs. 2014;74(12):1371-8. https://doi. org/10.1007/s40265-014-0260-2 PMid:25022953

 Adams D, Suhr OB, Hund E, Obici L, Tournev I, Campistol JM, et al. First European consensus for diagnosis, management, and treatment of transthyretin familial amyloid polyneuropathy. Curr Opin Neurol. 2016;29(Suppl 1):S14-26. https://doi.org/10.1097/ WCO.000000000000289

PMid:26734952

 Merlini G, Coelho T, Waddington Cruz M, Li H, Stewart M, Ebede B. Evaluation of mortality during long-term treatment with tafamidis for transthyretin amyloidosis with polyneuropathy: Clinical trial results up to 8.5 years. Neurol Ther. 2020;9(1):105-15. https://doi.org/10.1007/s40120-020-00180-w

PMid:32107748

- Adams D, Gonzalez-Duarte A, O'Riordan WD, Yang CC, Ueda M, Kristen AV, *et al.* Patisiran, an RNAi therapeutic, for hereditary transthyretin amyloidosis. N Engl J Med. 2018;379(1):11-21. https://doi.org/10.1056/NEJMoa1716153 PMid:29972753
- 17. Michalon A, Hagenbuch A, Huy C, Varela E, Combaluzier B, Damy T, *et al*. A human antibody selective for transthyretin amyloid removes cardiac amyloid through phagocytic immune cells. Nat Commun. 2021;12(1):3142. https://doi.org/10.1038/ s41467-021-23274-x

PMid:34035264

 Tschöpe C, Elsanhoury A. Treatment of transthyretin amyloid cardiomyopathy: The current options, the future, and the challenges. J Clin Med. 2022;11(8):2148. https://doi. org/10.3390/jcm11082148
PMid:35456241